 Vc/17-May-05
Week 18  starting 2 May 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  02 May 05 @ 5/2/2005 3:04:22 PM   
	3
	A1
	MBAL2126
	PT 12 Warm Up
	4.25

	12
	A2
	MBAL2135
	PT 13.2 Resist. Meas. 
	298.67

	2
	B1
	SSS090
	PT 5.2 1.5 KA Quench HF
	1.90

	5
	B2
	SSS171
	PT 12 Warm Up
	292.85

	10
	C1
	MBAR3325
	Prep 5 Cool Down
	297.59

	7
	C2
	MBAR3324
	Prep 5 Cool Down
	212.28

	11
	D1
	SSS126
	PT 14 Magnet MAPS
	297.41

	1
	D2
	SSS615
	PT 7 Magnet Trained
	1.90

	8
	E1
	MBAL2172
	PT 12 Warm Up
	267.38

	9
	E2
	MBAL2173
	Prep 5 Cool Down
	220.62

	4
	F1
	MBAL1179
	PTE Min. NRJ 3KA Diode Test
	1.91

	6
	F2
	MBBL2164
	Prep 5 Cool Down
	135.88


Coordinator this week: Marta; MTM support on call :   16-3602 
18  complete Magnet Tests cycles   ( 2171 ROBTC)
12  dipoles( 1 with MM)+ 6  SSS( 0  with MM) +  0  SSSS (with MM)
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :   J Halik
· General Points 
· AOB

Tests 

	Magnets under Test from 24-04-2005 to 30-04-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR3331 
	[image: image1.png]



	


	23-04-2005 
	0 h.
	21 h.
	23 h.
	16 h.
	19 h.
	26-04-2005 
	79 h.
	Stripping
	
	

	2
	TBA2
	MBBR3293 
	[image: image3.png]



	


	20-04-2005 
	34 h.
	25 h.
	43 h.
	16 h.
	12 h.
	25-04-2005 
	130 h.
	Stand By
	
	

	3
	TBB1
	SSS096 
	[image: image5.png]



	


	20-04-2005 
	26 h.
	26 h.
	18 h.
	9 h.
	28 h.
	25-04-2005 
	107 h.
	Stripping
	
	

	4
	TBB1
	SSS176 
	[image: image7.png]



	


	26-04-2005 
	35 h.
	11 h.
	26 h.
	9 h.
	16 h.
	30-04-2005 
	97 h.
	Stripping
	
	

	5
	TBB2
	SSS079 
	[image: image9.png]



	


	24-04-2005 
	27 h.
	13 h.
	26 h.
	8 h.
	16 h.
	27-04-2005 
	90 h.
	Stripping
	
	

	6
	TBC1
	MBBL1127 
	[image: image11.png]



	


	22-04-2005 
	23 h.
	27 h.
	38 h.
	14 h.
	14 h.
	26-04-2005 
	116 h.
	Stripping
	
	

	7
	TBC2
	MBBL1105 
	[image: image13.png]



	


	21-04-2005 
	21 h.
	23 h.
	35 h.
	14 h.
	24 h.
	26-04-2005 
	117 h.
	Stripping
	
	

	8
	TBC2
	MBBL1220 
	[image: image15.png]



	


	26-04-2005 
	43 h.
	24 h.
	13 h.
	14 h.
	8 h.
	30-04-2005 
	102 h.
	Stripping
	
	

	9
	TBD1
	SSS105 
	[image: image17.png]



	


	21-04-2005 
	22 h.
	25 h.
	16 h.
	7 h.
	18 h.
	25-04-2005 
	88 h.
	Stripping
	
	

	10
	TBD1
	SSS175 
	[image: image19.png]



	


	25-04-2005 
	25 h.
	11 h.
	16 h.
	7 h.
	12 h.
	28-04-2005 
	71 h.
	Stand By
	
	

	11
	TBD2
	SSS070 
	[image: image21.png]



	


	20-04-2005 
	31 h.
	12 h.
	26 h.
	7 h.
	10 h.
	25-04-2005 
	86 h.
	Stripping
	
	

	12
	TBE1
	MBBL2164 
	[image: image23.png]



	


	22-04-2005 
	0 h.
	24 h.
	8 h.
	16 h.
	14 h.
	24-04-2005 
	62 h.
	Stand By
	
	

	13
	TBE1
	MBBR3317 
	[image: image25.png]



	


	24-04-2005 
	22 h.
	20 h.
	14 h.
	12 h.
	17 h.
	28-04-2005 
	85 h.
	Stripping
	
	

	14
	TBE2
	MBAR3326 
	[image: image27.png]



	


	21-04-2005 
	25 h.
	39 h.
	18 h.
	13 h.
	12 h.
	25-04-2005 
	107 h.
	Stripping
	
	

	15
	TBE2
	MBBL1219 
	[image: image29.png]



	


	25-04-2005 
	46 h.
	25 h.
	17 h.
	16 h.
	15 h.
	30-04-2005 
	119 h.
	Stripping
	
	

	16
	TBF1
	MBBR3310 
	[image: image31.png]



	


	22-04-2005 
	22 h.
	22 h.
	22 h.
	14 h.
	20 h.
	27-04-2005 
	100 h.
	Stripping
	
	

	17
	TBF2
	MBAL2171 
	[image: image33.png]



	


	22-04-2005 
	23 h.
	29 h.
	45 h.
	-
	-
	27-04-2005 
	97 h.
	Not Found or Thermal cycle
	
	

	18
	TBF2
	MBAL2171 
	[image: image35.png]



	


	27-04-2005 
	0 h.
	15 h.
	11 h.
	15 h.
	6 h.
	29-04-2005 
	47 h.
	Stripping
	
	

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR3331
	12336.4 A.
(D1-U)
	12650 A.
(D2-U)
	[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBR3293
	10616 A.
(D1 - L)
	12512 A.
(D2-U)
	12827.5 A.
(D2-U-S01)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS096
	11333 A.
(QI-12)
	12319 A.
(QI-34)
	12789 A.
(QI-12)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS176
	11204 A.
(QI-12)
	11406 A.
(Q_E_12)
	11702 A.
(Q_E_34)
	12250 A.
(Q_I_12)
	12659 A.
(Q_E_12)
	12839.1 A.
(Qe-12)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-

	5
	SSS079
	11413 A.
(QE-34)
	11852 A.
(QI-34)
	12383 A.
(QI-34)
	12515.7 A.
(QI-34)
	12628 A.
(QI-34)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-

	6
	MBBL1127
	11665.7 A.
(D1-L)
	12269 A.
(D2 - U)
	12578 A.
(D1_U)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBL1105
	12189.2 A.
(D2-L)
	12404.3 A.
(D1-L)
	12572.9 A.
(D2-U)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBL1220
	11795 A.
(D2 - L)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS105
	12140 A.
(QI-34)
	12452 A.
(QE-34)
	12628 A.
(QI_12)
	[image: image45.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS175
	12344 A.
(QE-34)
	12620 A.
(QI-34)
	12688 A.
(QE-12)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS070
	11234.8 A.
(QI_34)
	11228 A.
(QE_12)
	11923 A.
(QE-34)
	12360 A.
(QI-34)
	12599 A.
(QI-12)
	12850 A.
[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-

	12
	MBBL2164
	11949 A.
(D1 - L)
	11537 A.
(D1 - L)
	[image: image48.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBR3317
	10577.1 A.
(D2-U)
	12303 A.
(D2-U)
	12333.1 A.
(D1-U)
	[image: image49.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAR3326
	11746 A.
(D1-U)
	11791 A.
(D1-U)
	12608 A.
(D2-L)
	12832 A.
(D2-U)
	[image: image50.png]



	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBL1219
	12518 A.
(D2_U)
	12630 A.
(D2_L)
	12794 A.
(D2_L)
	[image: image51.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBBR3310
	9130 A.
(D2-U)
	12073 A.
(D2-U)
	12007 A.
(D2-U)
	12391 A.
(D1-L)
	12525 A.
(D2-U)
	[image: image52.png]



	-
	-
	-
	-
	-
	-
	-

	17
	MBAL2171
	10169 A.
(D2-U)
	11326 A.
(D2_L)
	12034 A.
(D1-L)
	12254.6 A.
(D2-U)
	11749 A.
(D2-U)
	11839 A.
(D2-U)
	11849.2 A.
(D2-U)
	12388 A.
(D2_U)
	12630 A.
(D1_U)
	[image: image53.png]



	-
	-
	-

	18
	MBAL2171
	12162 A.
(D1-L)
	12390 A.
(D2_U)
	[image: image54.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


 

Week 17  :  24 April to 30 April 05  (18 Magnet Tests)
[   2171 On bench rapid thermal cycle   ]
[image: image55.png]Magnet Summary
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday _27_M


	TBD1


	Quench at 520 A during the powering of the EF corrector circuit of SSS175.
	ICS?
	
	Investigation by Giorgio. He suggested to open a non conformity because a bad clamping.

	Wednesday _27_M
	All
	Oracle problem: MTF stopped
	IT
	MTF
	

	Thursday _28_N
	TBA
	DAQ blocked. Resetting the crates is not sufficient.
	AB/CO
	Hf, lf
	A.Rijllart came and changed the MXI card and the cards in the crate.

	Thursday _28_N
	TBB1
	Connection of  voltage taps in the clamp for SSS176 is not good, detected by the corrector test at warm
	ICS
	
	Sleeve re-opened and reconnection done.

	Thursday _28_N
	TBD1
	One of the voltage tap wires for the main Quad is broken too far into the CFB to be re-soldered.
	ICS
	
	Maryline called Nikolay who asked to continue the connection without the VT.

	Thursday _28_N
	TBD2
	One of the voltage tap wires for the main Quad is broken too far into the CFB to be re-soldered.
	ICS
	
	Maryline called Nikolay who asked to continue the connection without the VT.

	Friday _29_M
	TBD1
	Permanent trigger because of configuration of bench corresponding to S4 
	MTM
	
	

	Friday _29_M
	TBE1
	Not MTF: cannot upload WP04
	IT
	MTF
	

	Friday _29_A
	TBA2
	IAP errors:  Q_loc offset
	MTM
	
	Maryline solved the problem by replacing the cable from the TRU to the rack.

	Friday _29_N
	TBA2
	Faulty quench on MB2135  trigger by potaim card 3 (Vbus EXT-Vbus INT)
	MTM
	
	

	Saturday_30_M
	TBF2
	Voltage during lyre test is 12.06V.
	MTM
	
	Maryline gave the OK to continue Voltage at warm acceptable as long as it stays stable during the test.

	Saturday_30_M
	TBC1
	MB3158: Data File ( L0504301147.ka01) not transferred to storage.


	AB/CO

MTM
	
	Maryline will transfer it Monday.



	Saturday_30_A
	TBF2
	Continuous message coming from the PLCSE. 
	MTM?
	
	Maryline advised to restart Tema, Server, PLC in this sequence.  

	Saturday_30_A
	TBF2
	HVrack MY92 -- connector D20 broken  

Lack of rack for HV tests.
	MTM
	
	

	Sunday_1_M
	TBB2
	Training could not be launched because of TCLaux_X2 and Cryo1_X2 signal not ok for PLCSE.
	ACR
	
	Problem rectified by the cryo-team.

	Sunday_1_M
	TBB2
	4 channels missing in hf DAQ
	AB/CO
	
	Maryline investigated the problem: one HF card faulty (to be confirmed)

	Sunday_1_A
	TBF1
	HV tests: Profibus is not working.
	]MTM
	
	Maryline disabled the interlocks signals coming from the PLCSE to the PLC of the HV rack until the repair Monday Morning.


 

 










